Author's response to reviews

Title: Psychosocial impact of perinatal loss among Muslim women

Authors:

ROSNAH SUTAN (rosnahsutan@ymail.com)
Hazlina miskam (razlinsha@gmail.com)

Version: 2 Date: 25 January 2012

Author's response to reviews: see over
Dear Editor,

Greeting from Malaysia

Thank you for considering our manuscript to be review by your expert editorial committee. The title of the manuscript is, Psychosocial impact of perinatal loss among Muslim women. I have edited the manuscript according to your suggestion. The corrections made are as follows:

Under methodology section:
Additional sentences were added at the end of the first paragraph as follow:
‘Permission to conduct the study was received from the UKMMC under specific grant FF-293-2008 and on-going survey was further support by grant UKM-HEJIM-komuniti-04-2010. Ethical approval was obtained from the UKMMC Research Ethics Committee and the study complied with the Helsinki Declaration. Written informed consent for participation in the study was obtained from the mothers who recently had perinatal loss’.

Under acknowledgment section:
‘Special thanks to Universiti Kebangsaan Malaysia who granted financial support to conduct this study (FF-293-2008 and UKM-HEJIM-komuniti-04-2010) and to the UKMMC Research Ethics Committee who approved it’.

I really hope that this manuscript has its potential to be accepted by your esteemed journal for publication.
Waiting for your kind response.

Thank you.

Warm regards
Assoc. Prof. Dr. Rosnah Sutan,
Community Health Department, Universiti Kebangsaan Malaysia Medical Centre, Bandar Tun Razak Cheras 56000 Kuala Lumpur. Tel no.:+603-91455900 email address: rosnahsutan@ymail.com or rosnah@medic.ukm.my Fax no.: +603-91456670.